Author’s response to reviews

Title: Short-term versus long-term psychotherapy for adult psychiatric disorders: a protocol for a systematic review with meta-analysis and Trial Sequential Analysis

Authors:

Sophie Juul (sophie.juul@regionh.dk)
Stig Poulsen (stig.poulsen@psy.ku.dk)
Susanne Lunn (susanne.luun@psy.ku.dk)
Per Sørensen (per.soerensen@regionh.dk)
Janus Jakobsen (jcj_ctu@icloud.com)
Sebastian Simonsen (sebastian.simonsen@regionh.dk)

Version: 2 Date: 03 Jul 2019

Author’s response to reviews:

Dear editor,

Thank you very much for taking your time to review our revised manuscript. We have complied to all your minor comments. Please see both our revised manuscript with track changes and the clean version, which are both uploaded. Further, we have also provided an updated PRISMA-P checklist.

On behalf of all the authors,

Sophie Juul, M.Sc.